**Author’s response to reviews**

**Title:** Knowledge of HIV/AIDS among older adults (50 years and above) in a peri-urban setting: A descriptive cross-sectional study

**Authors:**

REINDOLF ANOKYE (reindolfanokye@yahoo.com)

Enoch Acheampong (caposterl@yahoo.com)

Amy Budu-Ainooson (delonix.amy17@gmail.com)

Edmund Obeng (dreamsalivefoundation@gmail.com)

Emmanuel Tetteh (fishbone.emma@gmail.com)

Yvonne Acheampong (sambone344@gmail.com)

Nettey-Marbell Curtis (bpee.cm@gmail.com)

**Version:** 2  **Date:** 10 Oct 2019

**Author’s response to reviews:**

Negin paper included as suggested by Reviewer and can be found in the introduction section.

Studies on the subject available from other middle income countries like Ghana included as suggested by Reviewer and can be found in the introduction section.

HIV statistics being incident cases specified as suggested by Reviewer and can be found in the introduction section.

Details of translation included as suggested by reviewer and can be found in the Methods section.

Grammatical errors corrected throughout the manuscript

**REBUTTAL**

Please provide a reference for the closed-ended questions (page 6 - where are the Qs from?) - I have already indicated in the methods section that the questionnaire was designed in accordance to the study objectives by the investigators.